

Original Article



The development of a self-help intervention to build social confidence in people living with visible skin conditions or scars: a think-aloud study

Scars, Burns & Healing Volume 5: 1–8 DOI: 10.1177/2059513118822954 Article reuse guidelines: agepub.com/journals-permissions The Author(s) 2019 journals.sagepub.com/home/sbh

\$SAGE

Madeline Pasterfield^{1,3}, Sally-Anne Clarke² and Andrew R Thompson¹

Abstract

Introduction: People with a visible difference, such as scarring or a skin condition, can experience anxiety and intrusive reactions from others when in social situations. The use of products to conceal marks on the skin is provided in a number of different hospital services and by charities. However, there are relatively few psychosocial interventions available for these individuals.

Objectives: To examine the views of skin camouflage users and practitioners on the acceptability, usability and need for a specifically developed cognitive behavioural therapy (CBT) self-help booklet.

Methods: A think-aloud protocol and descriptive form of thematic analysis were used to ascertain participants' views of this novel psychosocial intervention. Nine participants took part in think-aloud interviews that were analysed using thematic analysis. Six skin camouflage users and three skin camouflage practitioners participated in the study.

Results: Support for the relevance, acceptability and usability of the booklet was found from both participants who used camouflage and those who provided it. However, some participants reported that they would envisage that some people would need additional support so as to be able to use the techniques described within the booklet.

Conclusions: This study represents an important step towards developing a brief self-help intervention for people with living with visible skin conditions or scars and demonstrates the importance of seeking feedback from experts by experience on theoretically informed psychological interventions for this patient group.

Keywords

Self-help, stigmatisation, scarring, skin conditions, think aloud, qualitative research

Corresponding author:

Andrew Thompson, Reader in Clinical Psychology, Department of Psychology, University of Sheffield, Cathedral Court, Sheffield, S1 1HD, UK.

Email: a.r.thompson@sheffield.ac.uk

¹Clinical Psychology Unit, Department of Psychology, University of Sheffield, Sheffield, UK

²The Royal Alexandra Children's Hospital, Brighton, UK

³North Bristol NHS Trust, Bristol, UK

Lay summary

Individuals with visible difference can experience anxiety, depression and decreased quality of life. However, there are relatively few psychosocial interventions available to help improve the lives of individuals with visible difference. This study provides preliminary evidence for the feasibility and usability of a novel psychosocial intervention, designed to improve social confidence and reduce social distress, for individuals with visible difference. Specifically, we sought to examine the views of skin camouflage users and practitioners on the acceptability, usability and need for a specifically developed cognitive behavioural therapy (CBT) self-help booklet. An interview which included a guide to facilitate reflection on the materials was used to ascertain participants' views of our novel psychosocial intervention. Nine participants took part, six skin camouflage users and three skin camouflage practitioners. The participants indicated that the self-help intervention had relevance, acceptability and usability. However, some participants reported that they would envisage that some people might need additional support to be able to use the techniques contained within the booklet. The findings from this study highlights the need for further development and testing of this psychosocial intervention, which holds promise for improving the lives of individuals with visible difference.

People living with visible skin conditions and scars can experience a variety of psychosocial difficulties including anxiety and depression¹ as well as increased levels of appearance specific distress^{2,3} that can adversely affect quality of life (QoL).^{4,5}

High levels of fear of negative evaluation and social anxiety have been found in people living with a range of conditions that affect appearance.^{6–8} Newell and Marks⁹ compared responses on a fear questionnaire for people with agoraphobia, social phobia conditions causing a visible difference. They found that people with a visible difference on their face showed similar patterns of avoidance to those with agoraphobia and social phobia. However, distress related to the perceived visibility of the condition is known to be poorly correlated with objectively rated disease severity and psychological variables have been argued to play a more significant role in adjustment.¹⁰

There are a number of theories that aim to understand the impact of living with a visible difference. 11 For example, Thompson and Kent 12 suggest that shame plays an important role and that the experience of social exclusion can lead people with a visible difference to develop social anxiety specific to their appearance concern. Models such as Kent's 13 posit that the experience of actual or perceived stigmatisation can lead to the development of appearance anxiety that is maintained with the use of avoidance and concealment. Other models have placed emphasis on the cognitive processes associated with overvaluation of appearance that may well be

associated with internalisation of cultural stereotypes or pressures.^{6,11,13–18}

There is good evidence for the effectiveness of cognitive behavioural therapy techniques (CBT) for the treatment of general social anxiety both when delivered face-to-face and in a self-help format.^{19,20} The potential utility of CBT-based interventions as a therapeutic tool for people with appearance altering conditions has been recognised⁶ and there has been some limited testing of self-help in this area.²¹⁻²³ CBT in this context would aim to alter unhelpful appearancerelated schema, challenge anxious thoughts related to negative social evaluation and remove safety behaviours such as subtle forms of concealment. However, such interventions also need to include strategies for managing episodes of actual stigmatisation. Kleve et al.24 reported upon the evaluation a small number of face-to-face CBT interventions. The intervention included assertiveness and social skills training as well as more typical CBT techniques, in a specialist disfigurement treatment centre and reported positive findings. Bessell et al.21 have also developed a computer-based intervention for the management of disfigurement-related distress. Nevertheless, reviews indicate that there remain relatively few self-help psychosocial interventions available for people specifically affected by skin conditions or with scarring; with the exception of the Kleve et al.²⁴ study, almost no evaluations of the effectiveness of face-to-face psychotherapeutic interventions are available in the literature. 25,26 Lavda et al. 25 conducted a meta-analysis that found that the majority of psychological interventions available

Pasterfield et al. 3

for skin conditions were simple behavioural interventions such as habit reversal. Nevertheless, they reported that such interventions showed promise and had a moderate effect on psychosocial outcomes. Further, a number of treatment manuals have been written to guide psychological practitioners; while some of these have been based on extensive research, the interventions themselves have yet to be fully tested.⁶

Investigating the potential benefits of psychological interventions with people specifically seeking camouflage treatment has the potential to provide a valuable opportunity for the development of low-intensity interventions that could be provided alongside camouflage but also be made available in primary mental healthcare and in dermatology clinics.

Whendevelopingacomplexpsychosocialintervention, the Medical Research Council (MRC) recommends the identification of the evidence base and appropriate theory related to the intervention, followed by formal assessment of feasibility and acceptability. ²⁷ Consequently, this study aims to explore the views of skin camouflage users and practitioners on the acceptability, usability and need for a CBT-based self-help booklet focusing on improving social confidence and reducing social distress.

Method

Design

A qualitative think-aloud protocol and semi-structured interview were used to ascertain the views of participants on the acceptability, usability and need for the self-help booklet. The 'think-aloud' technique gives access to the cognitive processes of the person and therefore provides rich verbal data.²⁸ This technique has been used in previous research developing healthcare interventions.^{29–32}

The self-help booklet being evaluated in this study was developed by the first and last author and is based on a theoretical model of social anxiety that has proven to be effective in guiding more intensive interventions.³³ The booklet includes components such as psycho-education, cognitive restructuring and graded exposure,³⁴ and acknowledges that people with a visible condition may experience actual intrusive reactions from others. The booklet was developed with feedback from providers and users of camouflage (see 'Acknowledgements'). It is available on the British Association of Dermatologists patient support website (www.skinsupport.org.uk).

Procedure

Recruitment. Camouflage users were recruited from a sample of patients who attended an NHS camouflage clinic and through the websites and social media sites of voluntary sector organisations associated with skin camouflage, skin conditions or burns. Ethical approval was obtained via the NHS research ethics approval system.

Participants were eligible to take part in the study if they were aged ≥ 16 years and currently using skin camouflage with a minimum of two months of continuous use. Participants were excluded if they were unable to participate in an interview in English. Additionally, people whose scars were a result of self-harm were not included, as such participants might be likely to have additional psychological issues that are less likely to be amenable to low-intensity self-help.³⁵

The study also sought to recruit a sample of camouflage practitioners. These participants were recruited nationally through a voluntary sector organisation. Potential participants were identified by the charity Changing Faces and an invitation email and information sheet were sent to them.

Data collection. Before the interview date, participants were posted a copy of the self-help booklet. A 'think-aloud' protocol and semi-structured interview were used.³⁶ During the interview, participants were asked to look through the self-help booklet section by section and speak out loud what they were thinking. Further structured questions on the booklet's content, layout, usability and utility were then asked at the end of each section. A standardised script was used with each participant.

The interviews were all completed face-to-face in the participant's home. Interviews were recorded using an encrypted digital recorder. In order to provide contextual background to the study, participants were asked to complete a set of demographic questions and two measures of psychological distress. Participants completed the Brief Fear of Negative Evaluation scale (BFNE),³⁷ a 12-item measure assessing a person's concern about the opinions of others, and the World Health Organization Quality of Life Brief Scale (WHOQoL Brief).³⁸

The camouflage practitioners took part in the think-aloud interview via telephone interviews that were recorded using an encrypted digital recorder. Recordings of the interviews were transcribed verbatim by an approved transcription

service who had signed a confidentiality agreement or by the first author. All transcripts were audited for accuracy.

Data analysis

The transcribed data were analysed using thematic analysis.³⁹ In thematic analysis, the data are examined for patterns, which are then defined and organised in order to encode the data.^{39,40} The analytic process included the following stages: familiarisation with the data; generation of initial codes; searching for themes; reviewing of themes; and defining and naming themes. This process was followed for each transcript separately and then transcripts were compared, contrasted and combined. NVivo⁴¹ was used to manage the data and facilitate analysis.

Results

Six camouflage users and three camouflage practitioners were recruited. A sample size of 8–10 participants is in line with other studies that have used this methodology to develop health interventions.²⁹ Table 1 gives a summary of the contextual information collected from camouflage-user participants. Practitioner participants were all women and had been offering skin camouflage services to clients for 1.5–10 years.

Analyses on the views of the camouflage-user participants will be presented first, followed by the findings from the practitioner participants.

Camouflage-user feedback

All of the camouflage users made positive comments that the information and techniques included appeared useful and relevant to them. However, some participants expressed concern that it might not be appropriate for everyone and some had reservations about providing a booklet without having other support available. Nevertheless, the majority described the booklet's content as helpful and relevant to people living with a visible skin condition:

I think the fact that you've got "my physical reactions" could actually provoke thoughts or how I feel or the behaviour, so that in the middle is quite relevant and obviously they all feed into each other don't they.' P5

All of the camouflage-user participants mentioned activities within the booklet that they viewed as being potentially helpful:

Table 1. Summary of background information from camouflage-user participants.

Camounage-user participants.	
Demographic	n
Age range (years)	49–72
Gender	
Female	6
Ethnic background	
Asian - Indian	1
White - British	4
White - Other	1
Reason for using camouflage	
Vitiligo	2
Skin condition (unspecified)	1
Birthmark	1
Scar from surgery	2
Length of time using camouflage	
Range	10 months – 40 years
Also a practitioner	2
BFNE score	
Range	14-60 (low-high)
WHOQoL score (higher scores denote better QoL, 100 maximum)	
Domain 1 range (Physical health)	19–100
Domain 2 range (Psychological)	38–94
Domain 3 range (Social relationships)	44–100
Domain 4 range (Environment)	63–100

BFNE, Brief Fear of Negative Evaluation Scale (Leary, 1983); WHOQoL, World Health Organisation Quality of Life – 26 (WHOQOL Group, 1998).

'[The thought challenging activity is] extremely important because this is where it tells you exactly what one is thinking and what to do with it.' P2

The booklet was reported as being understandable and clear. Five of the camouflage-user Pasterfield et al. 5

participants felt that the booklet was written in a clear and understandable way.

'It's very easy to understand.' P2

Some also made positive comments about the use of examples.

I think this one is clear and I think it's good to have the example here and then have the diagram for them.'P1

All of the camouflage-user participants described how they thought the booklet could or should be used in practice. Four of the camouflage-user participants stated that the booklet should not be completed without support.

I think when you're talking more about emotional support, you're talking about seeing someone face to face, not a booklet.' P5

Some participants said that the booklet could be used as a way of broaching the subject of the psychosocial impact with people coming to the skin camouflage clinic.

'Give you your options and say "this is what's available to you. What do you think you need?" P3

Others reiterated concern that additional support would need to be available within camouflage clinics if such a booklet was to be made available:

You have to remember that the camouflage practitioner is not a psychologist, to me this is something they should almost go to a counsellor.' P1

Five camouflage-user participants gave suggestions for changes or mentioned aspects that could be improved in the booklet. Three participants suggested that further examples would make some of the tasks clearer.

'Maybe an example in each so "my thoughts", "my feelings", "my reactions" and "my behaviours" so someone can see what you're thinking of and they can get on that wavelength.' P1

Another participant felt the length could be reduced and that the booklet should have a more personal style of writing.

'bit less wordy, it's... you're making it, I suppose in a way it has to grab people's attention and be personal I think.' P5

Five of the participants mentioned concerns about the 'Dealing with comments and reactions from others' section. Participants felt that some of the suggestions could be interpreted as confrontational and therefore should be amended.

'you need to get your responses a little bit more so it's giving them non-confrontational humorous ways to say something back.' P1

Camouflage practitioner feedback

All of the practitioners made positive comments about the self-help booklet.

1.1 Agree with the booklet's message. All practitioners agreed with the booklet's message; that people with a visible difference can struggle with social confidence.

'You know, they don't feel confident, I think that's a major, major factor and no matter how small the skin condition is.' C2

All three practitioners said the booklet was generally comprehensible, understandable and clear.

I think it is clear and you've kept it generally quite concise and that's all required I think.' C2

Overall, all practitioners thought the instructions given for the activities were clear.

I think that's fairly straightforward, I think that would be easy for someone to think "Yes, I think I could put down a couple of points." C1

All the practitioners felt that the majority of the information and activities included in the booklet were helpful and useful.

'Yes, I think that's a good exercise isn't it, because you've already written it all and you've gone through ticking thinking patterns.' C2

'I like the last paragraph about the eye contact... I think that's really good and also the fact that it's saying you're going to feel as though you're acting, but if you do it for long enough, you'll actually believe it and more importantly, so will other people.' C3

All the practitioners felt that the booklet could fit with skin camouflage services but

expressed a range of ideas about how. For example, whether all clients should be routinely sent a booklet before their appointment or whether it should be offered as an option during the session.

You either send it out to everybody and they come back or you wait until they come to clinic and if you feel that somebody is struggling on that side and you do get that, you can give them the booklet and you can perhaps talk through it with them.' C2

Although they felt the booklet was a useful tool, all practitioners said they did not think it would be relevant for all the people they see in camouflage clinics. Practitioners felt that the people more likely to use it would be those who are experiencing difficulties and want to build up their coping mechanisms.

'Some people, yes. I don't think everybody would want to do it but I think if somebody's really struggling it's good to have.' C2

Practitioner participants made suggestions for changes to the content of the booklet. For example, two practitioners felt that the introduction moved too quickly between discussing potential difficulties to highlighting stories of people who are coping well.

That little bit is slap bang in the middle of about people who are ok and whether you could bring it out rather than incorporate it ... have it separate..?'C2

Another felt that some of the suggestions in the section on dealing with the reactions of others could come across as confrontational.

I'm not sure I like the bits about "Thank you for staring, it's quite distracting"... I don't know... it's how it might come across... I'd just be worried that it would do the opposite of diffusing the situation...' C3

All participants suggested that the language used in the booklet could be changed in places.

Paragraphs like predicting the future and mind reading I thought you could perhaps move the emphasis from the negative to the positive. 'C3

A lack of clarity was highlighted within the instructions given for some activities.

T'm a bit confused here, so you write in alternative thoughts one to five, to what? To the above?'C1

Discussion

The aim of this study was to gain feedback on a CBT-based self-help booklet focusing on improving confidence in social situations. There was a large amount of agreement and shared opinion between camouflage users and practitioners leading to preliminary evidence for the acceptability, usability and need for such a booklet for people with skin conditions or scars. This also supports the potential utility of CBT models of social anxiety¹⁹ for this population. Generally, the information and activities included in the booklet were seen as helpful and appropriate and it was agreed that that the booklet could be a useful tool for at least some skin camouflage users. However, a caveat mentioned by many of the participants was that this type of material would not be relevant for all. A range of changes and improvements were suggested to the length, wording and instructions in order to make the booklet more readable and accessible to camouflage users. Section 5 of the booklet (dealing with comments and reactions from others) was highlighted by a number of participants as an area of concern. Some felt that certain responses could be interpreted as confrontational, suggesting further development and re-working of that section is necessary.

Overall, the participants felt there would be a place for such a booklet within services. The practitioners generally felt that it could be a useful tool within camouflage clinics. Some camouflage users agreed; however, others felt that depending on the level of emotional support needed by an individual, this work might be best completed within mental health services. There was a lack of consensus as to whether such a booklet should be posted to all camouflage clinic attendees as a matter of course, or whether it should be an option provided based on an expression of interest or need.

There are some limitations within our study that require consideration. The camouflage-user sample was purposively selected to be likely to have expert views that would be of use to the development of the intervention, nevertheless they are not fully representative of the population of people living with scars or who have skin conditions. Indeed, views were only obtained from specialist camouflage practitioners and service users, and not from other staff or patient populations where the intervention might be useful (e.g. within a dermatology clinic). Further, it is of particular importance when considering the transferability of our findings to be mindful that all of the participants were women and that none of

Pasterfield et al. 7

them had scars caused by traumatic injury or selfharm. Two participants within the sample had dual status as camouflage practitioners as well as being users of camouflage; while these participants have additional lived experience likely to have been of use in commenting upon the utility of the intervention, they are not representative of first-time users of camouflage services.

The first author was involved in the development of the intervention and also in collecting and analysing the data, and despite participants being encouraged to express criticism of the intervention, this nevertheless represents a risk bias. The intervention itself may have a number of limitations that warrant careful consideration ahead of preparation for further studies. For example, the written format of the intervention will place limitation on its use with certain populations. Clearly, further development of the intervention is needed and should be conducted in line with the MRC complex intervention development guidelines.²⁷ Further developmental work should also consider the cultural utility of the content intervention. Consequently, further usability and acceptability testing may be needed ahead of deciding whether or to conduct a randomised controlled trial (RCT). A RCT might also seek to investigate whether the intervention is more effective if completed with the guidance of a healthcare professional, as the existing literature on the use of self-help in treating anxiety suggests guided interventions are more likely to be effective³⁵.

The study suggests that camouflage use might in part be understood within a wider CBT framework of adjustment to visible difference.^{6,13} This suggests there is potential benefit in offering CBT-based interventions alongside camouflage services. Other variations of CBT, such as acceptance and commitment therapy, mindfulnessbased cognitive behavioural therapy compassion focused cognitive behavioural therapy, might also be useful to this population and warrant investigation in their own right. Indeed, a recent study indicates that some of the variables specifically targeted in these variations of CBT are associated with appearance-related anxiety in people with burn-related scarring.⁴²

This study is the first qualitative examination of the acceptability, usability and need for selfhelp for people with a skin conditions and scars. The study provides support for the utility of a CBT-based self-help booklet focussed on improving social confidence. Further development, feasibility and piloting work using the self-help intervention is now warranted.

Acknowledgements

Many thanks for providing preliminary pre-study feedback on the first draft of the self-help intervention booklet: Elizabeth Allen, Trustee and Principal Tutor British Association of Skin Camouflage (BASC); Kerry Montgomery, Senior Psychological Wellbeing Practitioner (IAPT), The Sheffield Health & Social Care NHS Foundation Trust; Krysia Saul, Camouflage Practitioner, Thanks are also due for assistance with recruitment to the study: Changing Faces; BASC; Helen Fletcher (Skin Camouflage Clinic, Northern General Hospital, Sheffield Teaching Hospitals); The Skin Camouflage Network.

Declaration of conflicting interests

The authors declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Funding

The author(s) disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: Funding to support publication in this journal was provided by The Katie Piper Foundation.

ORCID iD

Andrew R Thompson D https://orcid.org/0000-0001 -6788-7222



References

- 1. Dalgard FJ, Gieler U, Tomas-Aragones L, et al. The psychological burden of skin diseases: A cross-sectional multicenter study among dermatological out-patients in 13 European countries. [Invest Dermatol 2015; 135: 984-991.
- Brown B, Moss T, McGrouther D, et al. Skin scar pre-conceptions must be challenged: Importance of self-perception in skin scarring. J Plast Reconstr Aesthet Surg 2010; 63: 1022–1029.
- 3. Thompson AR, Clarke SA, Newell RJ, et al. Vitiligo linked to stigmatization in British South Asian women: A qualitative study of the experiences of living with vitiligo. Br J Dermatol 2010; 163: 481-486.
- 4. Green L. The effect of skin conditions on patients' quality of life. Nurs Stand 2010; 25: 48.
- 5. Schofield JG, Grindlay D and Williams H. Skin conditions in the UK: a health care needs assessment. Nottingham: Centre of Evidence Based Dermatology, University of Nottingham, 2009.
- 6. Clarke A, Thompson AR, Jenkinson E, et al. CBT for appearance anxiety: Psychosocial interventions for anxiety due to visible difference. Chichester: John Wiley & Sons, 2013.
- 7. Rumsey N, Clarke A, White P, et al. Altered body image: Appearance-related concerns of people with visible disfigurement. J Adv Nurs 2001; 48: 443-453.
- Rumsey N and Harcourt D. Body image and disfigurement: Issues and interventions. Body Image 2004; 1: 83-97.
- Newell R and Marks I. Phobic nature of social difficulty in facially disfigured people. Br J Psychiatry 2000; 176: 177–181.
- 10. Sampogna F, Sera F and Abeni D. Measures of clinical severity, quality of life, and psychological distress in patients with psoriasis: A cluster analysis. J Invest Dermatol 2004; 122: 602-607.
- 11. Thompson AR. Research directions in appearance research: Models for the future. In: Rumsey N and Harcourt D (eds)

Oxford Handbook of the Psychology of Appearance. Oxford: Oxford University Press, 2012: 91–109.

- Thompson A and Kent G. Adjusting to disfigurement: processes involved in dealing with being visibly different. Clin Psychol Rev 2001; 21: 663–682.
- Kent G. Testing a model of disfigurement: effects of a skin camouflage service on well-being and appearance anxiety. *Psychol Health* 2002; 17: 377–386.
- Cash TF. The influence of sociocultural factors on body image: Searching for constructs. Clin Psychol: Sci Pract 2005; 12: 438–442.
- Cash TF. Cognitive-behavioural perspectives on body image.
 In: Cash TF and Smolak L (ed) Body Image. London: The Guilford Press, 2011: 39–47.
- Moss T and Carr T. Understanding adjustment to disfigurement: The role of the self-concept. *Psychol Health* 2004; 19: 737–748.
- 17. Grandfield T, Thompson AR and Turpin G. An attitudinal study of responses to dermatitis using the implicit association test. *J Health Psychol* 2005; 10: 821–829.
- Moss TP, Lawson V, White P, et al. Salience and valence of appearance in a population with a visible difference of appearance: Direct and moderated relationships with self-consciousness, anxiety and depression. *PLoS One* 2014; 9: e88435.
- Clark DM. Implementing NICE guidelines for the psychological treatment of depression and anxiety disorders: the IAPT experience. *Int Rev Psychiatry* 2011; 23: 318–327.
- Haug T, Nordgreen T, Öst LG, et al. Self-help treatment of anxiety disorders: A meta-analysis and meta-regression of effects and potential moderators. Clin Psychol Rev 2012; 32: 425–445.
- Bessell A, Brough V, Clarke A, et al. Evaluation of the effectiveness of Face IT, a computer-based psychosocial intervention for disfigurement-related distress. *Psychol Health Med* 2012; 17: 565–577.
- Newell R and Clarke M. Evaluation of a self-help leaflet in treatment of social difficulties following facial disfigurement. *Int J Nurs Stud* 2000; 37: 381–388.
- Shah R, Hunt J, Webb TL, et al. Starting to develop self-help for social anxiety associated with vitiligo: Using clinical significance to measure the potential effectiveness of enhanced psychological self-help. *Br J Dermatol* 2014; 171: 332–337.
- Kleve L, Rumsey N, Wyn-Williams M, et al. The effectiveness of cognitive-behavioural interventions provided at Outlook: A disfigurement support unit. *J Eval Clin Pract* 2002; 8: 387–395.
- Lavda AC, Webb TL and Thompson AR. A meta-analysis of the effectiveness of psychological interventions for adults with skin conditions. *Br J Dermatol* 2012; 167: 970–979.
- Muftin Z and Thompson AR. A systematic review of self-help for disfigurement: Effectiveness, usability, and acceptability. *Body Image* 2013; 10: 442–450.
- 27. Craig P, Dieppe P, Macintyre S, et al. Developing and evaluating complex interventions: The new Medical Research Council guidance. *BMJ* 2008; 337: a1655.

- Fonteyn ME, Kuipers B and Grobe SJ. A description of think aloud method and protocol analysis. *Qual Health Res* 1993; 3: 430–441
- Protheroe J, Blakeman T, Bower P, et al. An intervention to promote patient participation and self-management in long term conditions: development and feasibility testing. BMC Health Serv Res 2010; 10: 206.
- Sadasivam RS, Delaughter K, Crenshaw K, et al. Development of an interactive, web-delivered system to increase providerpatient engagement in smoking cessation. *J Med Internet Res* 2011: 13: e87.
- Yardley L, Miller S, Teasdale E, et al. Using mixed methods to design a web-based behavioural intervention to reduce transmission of colds and flu. J Health Psychol 2011; 16: 353–364.
- Yardley L, Morrison LG, Andreou P, et al. Understanding reactions to an internet-delivered health-care intervention: Accommodating user preferences for information provision. BMC Med Inform Decis Mak 2010; 10: 52.
- Clark D and Wells A. A cognitive model of social phobia. In: Heimberg R, Liebowitz M, Hope DA, et al. (eds) Social Phobia: Diagnosis, Assessment and Treatment. New York, NY: Guilford Press, 1995: 69–93.
- Cuijpers P and Schuurmans J. Self-help interventions for anxiety disorders: An overview. Curr Psychiatry Rep 2007; 9: 284–290.
- Webster R, Thompson AR, Webb TL, et al. Self-help treatments and stepped care. In: Emmelkamp P and Ehring T (eds) The Wiley Handbook of Anxiety Disorders. London: Wiley, 2014: 1242–1255.
- Ericsson KA and Simon HA. Verbal reports as data. Psychol Rev 1980; 87: 215.
- Leary MR. A brief version of the Fear of Negative Evaluation Scale. Pers Soc Psychol Bull 1983; 3: 371–375.
- WHOQOL Group. Development of the World Health Organization WHOQOL-BREF quality of life assessment. Psychol Med 1998; 28: 551–558.
- Boyatzis R. Transforming qualitative information: Thematic analysis and code development. Thousand Oaks, CA: Sage, 1998.
- Braun V and Clarke V. Using thematic analysis in psychology. Qual Res Psychol 2006; 3: 77–101.
- 41. Nvivo 10 (Computer software). QSR International; 2012. Available at: http://www.qsrinternational.com.
- Shepherd L, Reynolds DP, Turner A, et al. The role of psychological flexibility in appearance anxiety in people who have experienced a visible burn injury. *Burns* (in press). Available at: eprints.whiterose.ac.uk/139531.

How to cite this article

Pasterfield M, Clarke SA and Thompson AR. The development of a self-help intervention to build social confidence in people living with visible skin conditions or scars: a think-aloud study. *Scars, Burns & Healing*, Volume 5, 2019. DOI: 10.1177/2059513118822954.